Author's response to reviews

Title: Near fatal Intoxication by Nicotine and Propylene Glycol Injection: A Case Report of an ELiquid Poisoning

Authors:

Mhedi Belkoniene (mhedi.belkoniene@gmail.com)

Jennifer Socquet (Jennifer.Socquet@gmail.com)

Denise Njemba-Freiburghaus (denise.njemba-freiburghaus@h-ne.ch)

Cyril Pellaton (Cyril.pellaton@h-ne.ch)

Version: 2 Date: 10 Feb 2019

Author's response to reviews:

Dear editor, Dear reviewers,

We take this opportunity to sincerely thank you for your comments that help us to significantly improve our manuscript. Please find below our last modifications suggested by the reviewer 1, before publication.

Reviewer 1:

Page 4 line 78 : We added two references (one from the US and one from the UK) including adult and paediatric population which confirm the raise of this new kind of intoxication worldwide.

Page 4 line 80 : We replaced the word "if" by "while" as suggested
Kind regards,

Mhedi Belkoniene, Jennifer Socquet Denise Njemba-Freiburghaus, MD and Cyril Pellaton, MD